Author's response to reviews

Title: Design of the Quality of Life in Motion (QLIM) study: a randomized Controlled Trial to evaluate the effectiveness and cost-effectiveness of a combined physical exercise and psychosocial training program to improve physical fitness in children with cancer.

Authors:

Katja I. Braam (katja.braam@vumc.nl)
Elisabeth M. van Dijk (em.vandijk@vumc.nl)
Margreet A. Veening (ma.veening@vumc.nl)
Marc B. Bierings (m.bierings@umcutrecht.nl)
Johannes H.M. Merks (j.h.merks@amc.uva.nl)
Martha A. Grootenhuis (m.a.grootenhuis@amc.uva.nl)
Mai J. Chin A Paw (m.chinapaw@vumc.nl)
Gerben Sinnema (g.sinnema@umcutrecht.nl)
Tim Takken (t.takken@umcutrecht.nl)
Jaap Huisman (drj.huisman@vumc.nl)
Gertjan J.L. Kaspers (gil.kaspers@vumc.nl)
Eline van Dulmen-den Broeder (e.vandulmen-denbroeder@vumc.nl)

Version: 3 Date: 10 November 2010

Author's response to reviews: see over
Dear Editor,

We changed the major and minor revisions such as you suggested.

1) We firstly changed the author list in the manuscript into First name, Initials, Last name.

2) We secondly included the two tables into the manuscript in a table section following the references and title and legend section of the figures.

3) Thirdly, we added the titles and legends of the figures in the manuscript and deleted them from the figure files.

4) Additionally, we added the full first name of Prof Gerben Sinnema in the email information section.

5) We deleted reference numbers in the legend of the second figure.


If you have any questions, we will be happy to answer them.

Yours sincerely

K.I. Braam
E.M. van Dijk